Author's response to reviews

Title: Recruitment of young women to a trial of chlamydia screening - as easy as it sounds?

Authors:

Helen Atherton (h.atherton@sgul.ac.uk)
Debbie Banks (deborahbanks@gmail.com)
Ruth Harbit (ruthharbit@cantab.net)
Linzie Long (m0500727@sgul.ac.uk)
Fiona Chadd (m0500446@sgul.ac.uk)
Phillip Hay (phay@sgul.ac.uk)
Sally Kerry (sgjdgp@sgul.ac.uk)
Ian Simms (ian.simms@hpa.org.uk)
Pippa Oakeshott (oakeshott@sgul.ac.uk)

Version: 2 Date: 15 June 2007

Author's response to reviews:

Dear Sirs

Please find attached a revised document containing the changes requested:

A title page.

Trial registration number at the end of the abstract.

Sections entitled: 'Competing Interests' and 'Authors Contributions'.

Legend for Figure 1 within the main text.

Separation of the conclusion from the discussion.

We hope you find these modifications acceptable.

Yours sincerely,

Helen Atherton